EO-1 Weekly Status Week of February 4 – February 17, 2010
Day of Year 035 - 048
Mission Day 3372 - 3385
Earth Observing One (EO-1) General.
· Scheduled 156 science Data Collection Events (DCEs) past week February 4-10.

· Scheduled 149 science Data Collection Events (DCEs) this week February 11-17.
This report covers a two week period due to the center being closed the first week due to heavy snow.
INSTRUMENTS

All instruments operated nominally this week 

· Performed instrument calibration

· Conducted ALI internal calibration (Type I) on February 8 at 01:19z

EO-1 Spacecraft Subsystems

Command and Data Handling (C&DH)

Continued to experience problems playing back engineering data from the Solid-State-Recorders.  Engineering data is being received during real-time contacts and for all science imaging events. 

Technology Activities

In addition to providing ongoing science data collection, the EO-1 extended mission supports on-orbit testbed activities for advanced technology and hyperspectral research.  The status of various validation efforts is contained in the following paragraphs.

SensorWeb & Virtual Observatory Demonstrations
On Thursday, February 4, there was an AIST 2008 SensorWeb 3G collaborators teleconference.  

Participants:  Dan Mandl, Vuong Ly, Rob Sohlberg, Jerry Hengemihle, and Joe Young.

Notes from this teleconference are as follows:

1.  Dan Mandl discussed the issue that, within the current funding, there are some activities that cannot be accomplished.  He is to identify such activities, together with required funding, and alert ESTO management.
2.  Status of the HyspIRI IPM Demo 1 was discussed particular with respect to what processors configuration will be utilized.
3.  Dan requested that charts for the ESTO review on March 1 be completed by February 22.
4.  After some delay, a cFE was finally delivered for integration on the Dryden Flight Research Center Global Hawk Unmanned Aerial System testbed. 
On Thursday, February 11, there was an AIST 2008 SensorWeb 3G collaborators teleconference.  

Participants:  Dan Mandl, Rob Sohlberg, Jerry Hengemihle, and Joe Young.

Notes from this teleconference are as follows:

1.  There was discussed the NASA Earth Observing Missions Applications Workshop that occurred on February 1-3 in Colorado Springs, CO. with respect to the workshop topics on Sensorwebs, automated workflows, onboard processing, and direct broadcast.
2.  Dan Mandl reported on his attendance at the ESTO AIST Technology Infusion Working Group meeting in Cocoa Beach, FL on February 9-11 and discussions he had with Steve Chien and Andrea Donnelan both of whom attended the above noted workshop in Colorado Springs.    The results of the NASA Earth Observation workshop in Colorado Springs were reported by Andrea and from that presentation it was apparent that SensorWeb capability provides key capabilities to demonstrate the societal benefits for the Decadal Survey Mission.  Dan is to send to the team the set of presentations from the ESTO meeting and a presentation on results from the workshop in Colorado Springs.
3.  Since there is a problem with getting support from Xilinx for the Spacecube 410 card, a decision was made to upgrade and buy a Spacecube 510 card.
On February 16, there was a Flood/Disease SensorWeb collaborators teleconference.  

Participants:  Dan Mandl, Stu Frye, Lenny Roytman, Serhiy Skakun, Pat Cappelaere, Joachim Post, Claudia Jobst, and Joe Young.

Notes from this teleconference are as follows: 

1.  There was a discussion on the desirability of producing a high resolution (1 m) digital terrain model of the Namibian flood basins via use of a LIDAR aerial survey.  The problem is that, while a high resolution terrain model is necessary to improve the flood forecasting, funding to make necessary flood model improvements so as to be able to accept the high resolution terrain model is not in hand.  Therefore, it is difficult to justify the $200k to $400k cost to conduct a LIDAR survey without improving the flood model.
2.  The most immediate need is to produce a project proposal for producing a flood/disease prediction model for Namibia that Guido van Langenhove can present to possible sources for financial support (e.g. World Bank).
3.  The White Paper can be used as a resource document to help produce the proposal.
4.  The proposal should address items from the Namibia PDNA (Post Disaster Needs Assessment) in addition to the White Paper.
5.  Since Guido could not participate in this teleconference, it was agreed to have another session the following day the 17th and focus on preparing the project proposal.
On February 16, the EO-1 Mission Science Office (MSO) conducted a meeting chaired by Betsy Middleton.  The participants were Betsy Middleton, Bruce Cook, Steve Ungar, Petya Campbell, Fred Huemmrich, Larry Corp, Lawrence Ong, Nathan Pollack, David Landis, Yen-Ben Cheng, Qingyuan Zhang, Dan Mandl, Stu Frye, and Joe Young.  

Topics most pertinent to EO-1 Operations that were discussed are as follows:

1.  Petya Campbell reviewed her Hyperion Cal/Val paper that is to be published in a Special Issue of the Canadian Journal of Remote Sensing.
2.  Petya gave report on the recent NASA Earth Observing Missions Applications Workshop where she co-chaired a session with Steve Chien on Rapid Access Applications.  At this meeting, Lawrence Friedl (NASA HQ) presented the same slides that were presented by Cheryl Yuhas at the recent AGU meeting which showed the Senior Review results where EO-1 was depicted as non-responsive to user requests.
3.  Bruce Cook was given an assignment to work with Stu Frye to develop and implement a plan to correct the above stated misconception about EO-1’s lack of responsiveness to user requests.
4.  Betsy Middleton requested that the sun-glint data collected by EO-1 be provided to Bo-Cai Gao and he be asked to correct the data.
5.  Dan Mandl reviewed the recent ESTO AIST Working Group meeting he attended.

6.  The next meeting will be on March 2.
On February 17, there was a Flood/Disease SensorWeb collaborators teleconference.  

Participants:  Dan Mandl, Stu Frye, Lenny Roytman, Guido van Langenhove, Joerg Szarzynski, Serhiy Skakun, Fritz Policelli, Joachim Post, Claudia Jobst, and Joe Young.

Notes from this teleconference are as follows: 

1.  Guido van Langenhove explained what the Namibian Flood/Disease Project Proposal should contain.
2.  The content of the proposal is to contain the following elements:
· Flood model update

· Improvement to website posting of accumulated precipitation and river gauge readouts by sub-catchment.

· High resolution digital terrain model from aerial LIDAR survey of flood plain regions.

· Radar, optical, and micro-wave observation sequences during pre and post flood as triggered by the SensorWeb.
· Rapid map production that combines historical averages with recent observation data.
· Other PDNA items.

3.  It was agreed that each contributor to the proposal submit their initial input prior to the next teleconference on February 23 according to the Work Package outline provided by the National Space Agency of Ukraine  and all the contributions be completed and integrated by UN-SPIDER prior to the teleconference on March 2.  Each element of the proposal will be one page that contains what is to be done, expected results, activity time schedule, and required resources with costing.
4.  Joerg Szarzynski offered use of the UN-SPIDER Knowledge Portal to post all Namibia related data including photos and videos taken during the late January trip to Namibia.
Dan Mandl, Stu Frye, Pat Cappelaere, and Jacqueline LeMoigne had a meeting at NASA Headquarters on February 17 with Michael Goodman, Karen Moe, Steve Berrick, and Martha Maiden to discuss the need for on-going support for the international SensorWeb work.  Cost estimates for needed support was provided.  Michael Goodman volunteered to act as a SensorWeb program scientist but stated he needed a 1-2 page white paper and an un-solicited proposal to permit him to take the next step.
Betsy Middleton, Fred Huemmrich, Fritz Policelli, and Dan Mandl attended an ISS Utilization Proposal meeting on February 17 where Harley Thronson described the process for center level review and prospects for award.  Evidently, funding for the ISS Utilization Proposals is $120M which is part of the President’s FY2011 budget request.
Autonomous Sciencecraft Experiment (ASE)

The ASE controlled EO-1 all week. 

EO-1 MISSION OPERATIONS CENTER

Real-Time

· Continued nominal spacecraft state of health

· Conducted 750-second inclination maneuver to maintain spacecraft MLT on February 9 at 18:40:41z

· Observed total average duty cycle of 60.7%, which is consistent with historical performance

· Computed observed thrust scale factor of 1.086 to be used as input to planning process for next maneuvers

· Conducted 750-second inclination maneuver to maintain spacecraft MLT on February 13 at 16:17:58z

· Changed planned maneuver from February 11 at 13:22:57z due to inclement weather and facility closings

· Observed total average duty cycle of 60.5%, which is consistent with historical performance

· Computed observed thrust scale factor of 1.072 to be used as input to planning process for next maneuvers

· Expect maneuvers to maintain descending node MLT at 10:00 am for approximately two months based on updated propagation models

· Planning to implement new code (CASPER r5.0.1) onboard spacecraft on February 23

· Awaiting to hear from stakeholders if any conflicts in schedule

· Inventorying open MOCRs to officially close any that are complete

Mission Planning

· Updating RTS files on SCP to match those currently in use on CMS for continued testing

Flight Dynamics
· Continuing troubleshooting of anomaly in PF1 tracking data that started January 19

· Observed no improvements in data during past week despite troubleshooting
Trending
· Observing and trending data from newly changed modified lunar calibration

· Comparing data from this modified lunar calibration to the last two calibrations to determine impact of removing the lamp calibrations on power consumption

System Administration
· C&A Audit

· Attended two audit meetings (toolbox and SA meeting)

· Scheduled and assisted the security team with the rescanning of the EO-1 Open IONet computers.

· The results report from the scan confirms that changes made by FOT corrected three medium vulnerabilities from the last scan. 

· The remaining medium and high vulnerabilities are on legacy systems that are scheduled to be updated/replaced as part of an existing POAM.

· Received the results report for the Flight Software Lab private network scan that took place in January.

· Two systems contained high vulnerabilities, but are Windows NT 4.0 which is no longer supported. Since they are on a private network the risk is mitigated but may still have to be accepted by system owner.

· Continued clearing space on /var partition of the legacy mission planning computer.

· Further investigated failed mail attempts to the EO-1 computer that is no longer active.

· Found three scripts that contained mail addresses on the inactive computer.

· Updated the scripts to replace all occurrences valid mail addresses.

· Discovered that mail aliases were being used by this computer 

· Updated the aliases file to replace all occurrences with valid mail addresses.

· Recompiled the mail aliases database.

· Improvements since the changes

· Files referencing failed mail attempts are no longer created in the mqueue directory.

· Log entries for the failed mail attempts no longer show up in the logfiles. Since our logs rotate, the large logs were eventually replaced smaller ones.

· This not only frees up space but also makes the logs easier to read.

· The /var partition is now at a safe capacity and is growing at a much slower rate than it was previously.

· Received notification that the /data partition on the legacy mission planning computer was getting full.

· This partition contains some oracle database files and index files.

· FOT ran their purge and defrag script to see if it would help to free up space.

· The script actually decreased the amount of space left on the partition.

· Contacted the Database Admin for help troubleshooting the problem.

· The DBA stated that our database requires manual intervention in order to grow in size. This means normal operations should be OK even though there is not much space left on the partition.

· It seems that running the script caused the index files to grow. This script should not be run again until the DBA has had time to troubleshoot and fix it.

· Multiple options were presented on how to free space on the /data partition. These options will be presented to the FOT.

· Started applying the latest FDCC security settings to the CNE computers. This process will continue until all of the CNE computers have been updated.

· Created two sets of weekly backups for the primary ASIST computers, new flight dynamics computers, ASPEN computers, countdown clock, new data processing computers, and MOC CNE computer.

GROUND AND SPACE NETWORK

Station Downtimes

· Observed no station down times
Operational Discrepancies
· This weekly is from past two weeks due to inclement weather and facility closings

· Due to inclement weather and facility closings, conducted inclination maneuver on Saturday, February 13 rather than February 11.

· Continued anomaly in PF1 tracking data that started January 19

· Removed PF1 tracking data manually from flight dynamics processing

UPCOMING EVENTS

· Institute ASE software update onboard spacecraft to correct for leap second issue, single bit error issue, and temperature log size issue in ASE software that is scheduled for no earlier than February 23
Imagery Status

Scenes and Engineering Cals planned for weeks of February 4 – February 17, 2010      156 & 149
Total scenes and engineering calibrations planned for entire mission – approximately 51,518
Total Scenes:  ALI scenes in the Level 0 archive              47,080             (as of February 17, 2010)

                         Hyperion scenes in the Level 0 archive    46,832             
Publications and Presentations Status 

404 publications 

284 external presentations

53 articles and press releases
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